Author’s response to reviews

Title: Progression of coronary atherosclerotic plaque burden and relationship with adverse cardiovascular event in asymptomatic diabetic patients

Authors:

junjie yang (yangjj301@gmail.com)

Guanhua Dou (109310347@qq.com)

Christian Tesche (tesche.christian@googlemail.com)

Carlo N. De Cecco (dececco@musc.edu)

Brian E. Jacobs (jacobsbr@musc.edu)

U. Joseph Schoepf (schoepf@musc.edu)

yundai chen (cyundai301@sina.com)

Version: 4 Date: 26 Dec 2018

Author’s response to reviews:

1. Please provide a list of three to ten key words after the abstract.

Respond: We have provided key words as required.

2. Thank you for providing an Availability of data and materials subheading in your Declarations section. Please describe where the raw data generated/analysed during your study can be found, if any is unaccounted for. If you wish to share the data through correspondence then please include the statement “The datasets used and/or analysed during the current study are available from the corresponding author on reasonable request.” If you do not wish to share your data, please state this along with the reason.

Respond: We wish to share our data and included the statement as required.

3. In the Funding section, please also describe the role of the funding body in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.

Respond: We described the role of funding body in the Funding section as required.
4. Please note that based on the current authors’ contributions section, several listed authors do not automatically qualify for authorship.

We ask that you ensure that author contributions are in line with the ICMJE guidelines (below), and that all listed authors have performed all four points specified below.

An ‘author’ is generally considered to be someone who has made substantive intellectual contributions to a published study. According to the ICMJE guidelines, to qualify as an author one should have:

a) made substantial contributions to conception and design, or acquisition of data, or analysis and interpretation of data; AND
b) been involved in drafting the manuscript or revising it critically for important intellectual content; AND
c) given final approval of the version to be published. Each author should have participated sufficiently in the work to take public responsibility for appropriate portions of the content; AND
d) agreed to be accountable for all aspects of the work in ensuring that questions related to the accuracy or integrity of any part of the work are appropriately investigated and resolved.

Anyone listed as an author must be included in this section. If you choose to change your author list you will need to fill out a change in authorship form and send it by email to the Editorial office to be approved by the Editor. The form can be found here: https://www.biomedcentral.com/getpublished/editorial-policies#authorship.

Anyone who contributed towards the article who does not meet the criteria for authorship can be acknowledged in the ‘Acknowledgements’ section.

Respond: Much appreciated to Editor comments. We ensured that all listed authors have performed all four points specified above.